

# **Earth's Future**

# COMMENTARY

10.1029/2023EF003535

#### **Key Points:**

- The coastal planetary boundary layer (PBL) is highly dynamic and poses challenges to accurately observe and model air quality (AQ), water quality, and ocean color
- Coastal AQ observations and simulations via field campaigns and networks have advanced our understanding of processes coupled by the PBL
- Extensive observations of the coastal PBL are needed to achieve interdisciplinary research goals at the land/water interface

#### **Correspondence to:**

J. T. Sullivan, john.t.sullivan@nasa.gov

#### Citation:

Sullivan, J. T., Stauffer, R. M., Thompson, A. M., Tzortziou, M. A., Loughner, C. P., Jordan, C. E., & Santanello, J. A. (2023). Surf, turf, and above the Earth: Unmet needs for coastal air quality science in the planetary boundary layer (PBL). *Earth's Future*, 11, e2023EF003535. https://doi. org/10.1029/2023EF003535

Received 24 JAN 2023 Accepted 21 MAY 2023

#### **Author Contributions:**

**Conceptualization:** Ryan M. Stauffer, Carolyn E. Jordan **Investigation:** Maria A. Tzortziou

Methodology: Ryan M. Stauffer, Maria A. Tzortziou, Christopher P. Loughner, Carolyn E. Jordan, Joseph A. Santanello Visualization: Ryan M. Stauffer Writing – original draft: Ryan M. Stauffer

Writing – review & editing: Ryan M. Stauffer, Anne M. Thompson, Maria A. Tzortziou, Christopher P. Loughner, Carolyn E. Jordan, Joseph A. Santanello

© 2023 The Authors. This article has been contributed to by U.S. Government employees and their work is in the public domain in the USA.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# Surf, Turf, and Above the Earth: Unmet Needs for Coastal Air Quality Science in the Planetary Boundary Layer (PBL)

John T. Sullivan<sup>1</sup>, Ryan M. Stauffer<sup>1</sup>, Anne M. Thompson<sup>1,2</sup>, Maria A. Tzortziou<sup>1,3</sup>, Christopher P. Loughner<sup>4</sup>, Carolyn E. Jordan<sup>5,6</sup>, and Joseph A. Santanello<sup>1</sup>

<sup>1</sup>NASA Goddard Space Flight Center, Greenbelt, MD, USA, <sup>2</sup>Goddard Earth Sciences Technology and Research (GESTAR) II, University of Maryland Baltimore County, Baltimore, MD, USA, <sup>3</sup>Earth and Atmospheric Sciences, City University of New York, New York, NY, USA, <sup>4</sup>Air Resources Laboratory, National Oceanic and Atmospheric Administration, College Park, MD, USA, <sup>5</sup>National Institute of Aerospace, Hampton, VA, USA, <sup>6</sup>NASA Langley Research Center, Hampton, VA, USA

Abstract Coastal areas are some of the most densely populated and economically important regions in the world. As such, protecting the health of the human population and ecosystems at the coastal interface and understanding the impacts of environmental stressors such as air pollutants provides wide-ranging benefits. Air quality (AQ) processes within coastal regions have been studied using ground and space-based platforms, with intensive field campaigns focused on addressing key science questions that are typically partitioned into either direct atmospheric effects (e.g., anthropogenic emissions creating air pollution) or indirect processes and feedback loops (e.g., terrestrial/marine biogenic processes modifying atmospheric properties). The atmospheric planetary boundary layer (PBL) and its depth (or height) connect land, air, and the water surface via many pathways, especially with transport and exchange processes tied to the complexities of the coastal interface. We still cannot accurately characterize-through field, aircraft, or space-based observations-the spatial and temporal PBL variability and processes within the PBL that couple together coastal dynamics and air quality. Several upcoming geostationary and polar-orbiting satellite missions are likely to make significant progress in characterizing these air/land/water interactions over the next decade. Here, we present a framework of the current understanding of the PBL's role in coastal regions, primarily regarding air quality and atmospheric deposition, to motivate future concerted efforts from ground- and space-based platforms to achieve a holistic understanding of the coastal interface.

**Plain Language Summary** Coastal areas are some of the most densely populated and economically important regions in the world. Protecting the health of the human population and ecosystems at the coastal interface and understanding the impacts of key environmental stressors, such as air pollution, provides wide-ranging benefits. The atmospheric planetary boundary layer (PBL) connects land, air, and the water surface, yet there are remaining challenges in accurately characterizing the role of the PBL in coastal areas and its variability. Here, we present a framework of the current understanding of the PBL's role in coastal regions, primarily regarding air pollution, to motivate future concerted efforts from ground- and space-based platforms to achieve a holistic understanding of the coastal interface.

#### 1. Introduction

Most of the world's megacities are in the coastal zone (Brown et al., 2013), and nearly 50% of the US population lives in coastal counties (Crossett et al., 2004), facing urban environmental challenges that occur within the complex coastal atmospheric planetary boundary layer (PBL). While the definitions of the PBL and the PBL height/depth (PBLH) vary depending on scientific application (e.g., gradients in temperature/atmospheric stability, composition, or turbulence), we generally refer to the PBL as the lowest layer of the atmosphere that is directly affected by heat and moisture fluxes and friction from the Earth's surface (Medeiros et al., 2005). The interaction of both land and ocean surfaces with the PBL is a very active area of research, with distinct communities (land-atmosphere and ocean-atmosphere) and driving science questions (e.g., Santanello et al., 2018 and refs therein; Teixeira et al., 2021). Routine measurements of the PBLH are provided via high temporal and vertical resolution ground-based measurements, such as radiosonde soundings, commercial aircraft measurements (e.g., Aircraft Meteorological Data Relay (AMDAR), Y. Zhang et al., 2020), and lidar backscatter/ceilometer. For near global coverage, satellite estimates of PBLH can be derived from spaceborne lidar instruments, such as the



# The PBL Controls Coastal Dynamics and Pollution Episodes

Figure 1. Conceptual model of the processes impacting the planetary boundary layer at the coastal interface.

Cloud-Aerosol Lidar and Infrared Pathfinder Satellite Observations (CALIPSO), vertical sounders, such as the Atmospheric Infrared Sounder (AIRS), and GPS radio occultation, such as the GPS Occultation Analysis System (GOAS) and Global Navigation Satellite System (GNSS). Unfortunately, current spaceborne PBLH estimates have significant limitations with vertical or temporal coverage and generally only monthly/seasonally averaged products are considered reliable for scientific analysis. There is no single instrument that provides daily global coverage of accurate PBLH measurements, which requires these platforms and their retrievals to be used in conjunction with atmospheric model simulations and ground-based evaluation to achieve a better understanding of PBLH variability.

Current observational platforms have strengths and weaknesses related to spatial coverage and temporal frequency of observations and accuracy of the measurements. Spaceborne PBLH measurements are particularly challenging in coastal areas given the large horizontal gradients in terrain, surface albedo, and emissivity, along with the high likelihood that these gradients lie within a single satellite ground pixel. Ground-based measurements have high temporal and vertical resolution but are not able to provide a true global picture. In addition, the ability to "resolve" the PBL from satellite is inherently limited as the lower troposphere is the farthest from space and have reduced sensitivity to the PBL which can be confounded by clouds and aerosols.

These observational limitations in the PBL present a challenge to the scientific community who are trying to better understand its sub-hourly chemical and dynamical traits. The PBL and its depth are directly responsible for the temporal, vertical, and spatial scales on which many environmental interactions take place (Figure 1). In situations where a shallow PBL depth is formed, atmospheric pollutants emitted from the surface are trapped within this shallow layer. In contrast, as the PBL depth grows convectively during daytime, pollutants are diluted through vertical transport and mixing. Convective mixing of the PBL can also entrain pollutants transported at higher altitudes from upwind regions, adding to the budget of boundary layer pollutants. Therefore, to better understand coastal AQ science, it is imperative to better understand the controlling factors (e.g., meteorology, atmospheric composition and transformation, terrain, emissions, wet and dry deposition) of the coastal environment and how they influence and are influenced by changes in the PBL (Crosman & Horel, 2010).

Changes in environmental conditions often occur at varying time scales, ranging from quickly evolving pollution events (e.g., J. Zhang et al., 2020) to decadal changes in de/nitrification of bodies of water (e.g., Clune et al., 2021). Therefore, many researchers have naturally focused on a specific coastal environmental issue addressed through a disciplinary perspective and assumptions. However, the multifaceted impacts of PBL on coastal processes demand that current and future observations, simulations/assimilations, and funding opportunities for the Earth System shift focus beyond one discipline toward understanding PBL evolution as an integrated and interconnected system.



#### Table 1

A Summary of Various Collaboratively Supported Field Campaigns, Primarily in the U.S. and North America, Examining Air and Water Quality Near the Coastal Interface

	Campaign	Time period	Focus region	Reference(S)
1	CBODAQ/DISCOVER-AQ <sup>a</sup>	July 2011	Mid-Atlantic US	Goldberg et al. (2014)
2	GoMex/DISCOVER-AQ <sup>b</sup>	September 2013	Gulf of Mexico	Kowalewski and Janz (2014)
3	DANCE°	July-August 2014	Mid-Atlantic US	D. K. Martins et al. (2016)
4	NAAMES <sup>d</sup>	2015–2018	North Atlantic/Canada	Behrenfeld et al. (2019)
5	KORUS-AQ/OC <sup>€</sup>	May–June 2016	Korean Peninsula	C. E. Jordan et al. (2021a, 2021b), Tzortziou et al. (2018)
6	OWLETS 1–2 <sup>f</sup>	June–July 2017–2018	Mid-Atlantic US	Sullivan et al. (2019), Dreessen et al. (2023),
7	LMOS <sup>g</sup>	May–June 2017	Lake Michigan	Stanier et al. (2021)
8	LISTOS <sup>h</sup>	June–September 2018	NYC/Long Island Sound	Karambelas (2020)
9	SCOAPE <sup>i</sup>	May 2019	Louisiana/GOM	Thompson et al. (2020, 2023)
10	TRACER-AQ <sup>i</sup>	September 2021	Houston/GOM	Jensen et al. (2022), Judd et al. (2021)

<sup>a</sup>GEO-CAPE Chesapeake Bay Oceanographic campaign with DISCOVER-AQ (Baltimore, MD/Washington D.C. 2011 Deployment); https://earth.gsfc.nasa.gov/ocean/ campaigns/geo-cape-chesapeake-bay-oceanographic-campaign-discover-aq-cbodaq. <sup>b</sup>Gulf of Mexico field campaign/Deriving Information on Surface conditions from Column and Vertically Resolved Observations Relevant to Air Quality (Houston, TX 2013 Deployment); https://www-air.larc.nasa.gov/missions/discover-aq/ discover-aq.html. <sup>c</sup>Deposition of Atmospheric Nitrogen to Coastal Ecosystems experiment; https://sites.psu.edu/dance2014/. <sup>d</sup>North Atlantic Aerosol and Marine Ecosystem Study; https://science.larc.nasa.gov/NAAMES/. <sup>e</sup>Korea-United States Ocean Color experiment; https://www-air.larc.nasa.gov/missions/korus-aq/. <sup>f</sup>Ozone Water-Land Environmental Transition Study; https://www-air.larc.nasa.gov/missions/owlets/. <sup>g</sup>Lake Michigan Ozone Study; https://www-air.larc.nasa.gov/missions/ Imos/. <sup>h</sup>Long Island Sound Tropospheric Ozone Study; https://www-air.larc.nasa.gov/missions/listos/index.html. <sup>i</sup>Satellite Coastal and Oceanic Atmospheric Pollution Experiment; https://www-air.larc.nasa.gov/missions/scoape/index.html. <sup>i</sup>Tracking Aerosol Convection Interactions Experiment—Air Quality; https://www-air.larc. nasa.gov/missions/tracer-aq/.

To better meet these interdisciplinary challenges occurring within the PBL, many regional sub-orbital (i.e., ground-based, ship-based, and airborne) intensive experiments have been carried out jointly with state and federal agencies geared toward improving the lives of their stakeholders (see Table 1). As the AQ community enters the era of geostationary satellite observations, coordination of additional campaigns and enhancement of existing observing networks (e.g., NASA's Tropospheric Ozone Lidar Network (TOLNet), the Unified Ceilometer Network, NASA's/ESA's Pandonia Global Network (PGN), NASA MicroPulse Lidar NETwork (MPLNET), and routine radiosonde/ozonesonde launches) offers an opportunity to quantify PBL dynamics and improve the representativeness of spaceborne measurements of PBL processes.

The 2017 National Academies Decadal Survey (*Thriving On Our Changing Planet*, NASEM, 2018), specifically proposed a combination of geostationary and polar orbiting satellites, airborne platforms, and ground-based networks to characterize the strong diurnal cycles of PBL processes. Cross-discipline studies and working groups have been established to aid in PBL characterization, such as through NASA's PBL Incubation Study Report (Teixeira et al., 2021; https://science.nasa.gov/science-pink/s3fs-public/atoms/files/NASA\_PBL\_Incubation\_Final\_Report\_2.pdf). While NASA's Decadal Survey Incubation activities are heavily focused on spaceborne applications, a necessary component is incorporating findings and methods from prior sub-orbital campaigns, illustrating their advances in PBL science in coastal areas. Here, we synthesize findings from previous ground-based/airborne campaign measurements to address the interdisciplinary nature of existing research questions relevant to PBL processes and requirements of future measurement strategies. Specific questions addressed are:

- What have we collectively learned from coastal AQ observations and simulations (through field campaign efforts or observation networks) that advance our understanding of processes coupled by the PBL? (Section 2)
- How representative are current and future measurements from space of PBL characteristics and the feedback/ exchange processes impacting AQ? (Section 3)
- More generally: How are research goals surrounding the land/water interface prime examples of the need to increase the interdisciplinary nature of atmospheric scientific research? (Section 4)







**Figure 2.** OMI version 4 tropospheric column NO<sub>2</sub> average in Dobson Units (DU) for 2019 (https://avdc.gsfc.nasa.gov/pub/data/satellite/Aura/OMI/V03/L3/OMNO2d\_HR/OMNO2d\_HRM/). Locations of campaigns listed in Table 1 (except KORUS-OC; 5) are shown. Black outlined regions indicate current U.S. EPA-designated Ozone Non-Attainment Areas based on the 2015 National Ambient Air Quality Standards (NAAQS; https://www.epa.gov/green-book/green-book-gis-download; Last Updated 1 February 2022).

In this paper, our primary focus is on the complexities of the coastal PBL and the impact they have on AQ, deposition of pollutants, and coastal ecological processes. Our goals are to provide more context and insight on the above questions. This is accomplished by defining important land/water feedback processes in the PBL, discussing measurement strategies employed to better understand these processes, and identifying critical measurement gaps of PBL characteristics that must be addressed to answer important interdisciplinary science questions.

## 2. Lessons From Recent Coastal Field Campaigns

## 2.1. Summary of Recent Campaigns

The coastal PBL facilitates interactions between anthropogenic and biogenic pollutants (both their sources and sinks) that contribute to poor AQ (as illustrated in Figure 1). We draw from past NASA and partnering organization supported field campaigns (listed in Table 1 and shown on Figure 2) that sought to address some of the scientific questions listed above. These campaigns took place generally in summertime in and near various bodies of water within the United States since 2011. An exception is the KORUS-OC (Korea-U.S. Ocean Color experiment) that occurred in coastal waters

surrounding the Korean peninsula in 2016 (C. E. Jordan et al., 2021a, 2021b; Thompson et al., 2019; Tzortziou et al., 2018). KORUS-OC is instructive because it combined air and water quality measurements from research vessels in conjunction with the comprehensive KORUS-AQ (Korea-U.S. Air quality) campaign that used observations from aircraft, ground sites, and satellites (Crawford et al., 2021). Measurements during KORUS OC/AQ were used to examine transport of atmospheric pollutants, including urban nitrogen pollution, over the coastal ocean and their impacts on atmospheric correction of satellite ocean color and retrievals of coastal ecological processes. The coordinated KORUS OC/AQ fieldwork program is an excellent model for future interdisciplinary land/water interface science. For example, Tzortziou et al. (2018) measured total column NO<sub>2</sub> (TCNO<sub>3</sub>) and  $O_3$  (TCO<sub>3</sub>) over coastal waters around the Korean peninsula using shipborne Pandora instruments as part of KORUS-OC. These observations were integrated with KORUS-AO observations and data products such as ground-based observations at coastal land sites, synoptic satellite imagery, and air mass trajectory simulations. They reported that variability in TCO<sub>3</sub> over the coastal ocean was relatively small (20%) and mostly exhibited a quasi bi-weekly oscillation driven primarily by larger scale meteorological processes and synoptic weather fronts that were captured successfully by AURA-OMI. TCNO2, however, exhibited small-scale and short-term variability by more than an order of magnitude, linked to urban nitrogen pollution. If not properly accounted for in atmospheric correction retrievals of coastal ocean color, such variability in coastal atmospheric composition will result in a false spatial or temporal variability in ocean biogeochemical properties and ecological processes.

The U.S.-based campaigns have focused on the Mid-Atlantic and North Atlantic Ocean (see Table 1 for acronyms; DISCOVER-AQ, CBODAQ, DANCE, NAAMES, OWLETS, LISTOS), Gulf of Mexico (DISCOVER-AQ, GoMex, SCOAPE, TRACER-AQ), and Lake Michigan (LMOS). These areas were targeted primarily because of their proximity to non-attainment regions for the U.S. EPA's National Ambient Air Quality Standard (NAAQS) for ozone (https://www.epa.gov/green-book), and the abundance of local anthropogenic emissions (NO<sub>2</sub> and ozone non-attainment areas shown on Figure 2). Our focus is primarily on the U.S. and North America given that there are several upcoming polar-orbiting and geostationary satellite missions targeting this region that will aid our understanding of the coastal PBL.

#### 2.2. Characterizing Processes Driving PBLH Evolution

Most recent coastal AQ-focused campaigns (Table 1) highlighted examples of differential heating between the land and water, which impacts the PBL in specific ways that are more complex than traditional continental regions. This gradient affects the height or mixing depth of the PBL and causes winds to stagnate or reverse flow directions during the daytime or transitional periods, such as sunrise or sunset. Differential surface heating and cooling between land and water can also lead to a gradient in PBL depth at the coastline resulting in large





**Figure 3.** Coincident ozonesonde profiles from 1 July 2018 during the OWLETS-2 campaign in the northern Chesapeake Bay region. (a) Ozone mixing ratio (thick, solid lines) and (b) Wind direction profiles up to 3 km AMSL. The three ozonesondes were launched within minutes of each other (1715 UTC; 1215 LST). The map (c) shows the locations of the three ozonesonde launch locations using the same color scheme as the vertical profiles. Note HMI is 30 km due east of UMBC and HUBV is 26 km SW of UMBC. Time series of ozone lidar profiles at (d) UMBC and (e) HMI are also shown with an overlay of ozonesonde profiles (blue/green triangles) and surface ozone analyzer data (bottom).

gradients in cloud coverage, air pollution concentrations, and dry deposition (Dreessen et al., 2023; Goldberg et al., 2014; Loughner et al., 2016; Mazzuca et al., 2019; Stauffer & Thompson, 2015; Sullivan et al., 2019).

Under weak synoptic scale forcings, differential surface heating or cooling between the land and water can lead to localized circulation patterns, such as sea, bay, or lake breezes (Miller et al., 2003). Stagnation in the wind and a flow reversal associated with localized onshore or offshore breezes confines fresh emissions (Dacic et al., 2020; Kotsakis et al., 2022; Loughner et al., 2011, 2014; Stauffer et al., 2015) and the accumulation of pollutants that would otherwise be transported beyond the urban region. These conditions contribute to ozone non-attainment (see Figure 2) designations, whether in populated U.S. coastal areas, such as the Gulf Coast (e.g., Houston, New Orleans), the Chesapeake Bay, Lake Michigan, and Long Island Sound regions (Goldberg et al., 2014; Loughner et al., 2014; Nauth, 2021; Stauffer et al., 2015), or in coastal megacities worldwide, such as Seoul (Peterson et al., 2019). Such meteorological conditions near New York City were characterized by particularly low-speed southerly/westerly winds and were previously shown to drive high NO<sub>2</sub> pollution episodes even under extreme reductions in emissions from the transportation sector during the COVID-19 pandemic (Tzortziou et al., 2022). Furthermore, in many occurrences a stable PBL leads to fair weather cumulus clouds at the top of the PBL over land but not over water causing contrasting chemical processing regimes (i.e., increased ozone production efficiency over water) that may occur directly in the shallow PBL (often 20–60 m AGL) within the marine environment versus those in a traditional land environment that have much deeper PBL depths (illustrated in Figure 1).

#### 2.3. Campaign Example in the Chesapeake Bay (OWLETS-2)

Figure 3 illustrates the complexity of land/water PBL gradients that lead to differing vertical distributions of pollutants. During the Chesapeake Bay region 2018 OWLETS-2 campaign (Table 1), three ozonesondes were launched within minutes of each other at three locations: within the Chesapeake Bay (Hart Miller Island; HMI), near Baltimore (University of Maryland Baltimore County; UMBC), and farther inland (Howard University

Beltsville Campus; HUBV). Time series of ozone lidar profiles at (d) UMBC and (e) HMI are shown with an overlay of ozonesonde profiles (blue/green triangles corresponding to the ozonesonde color) and surface ozone analyzer data.

The ozone sonde profiles represent a "local snapshot" of the gradients in near-surface ozone pollution, vertical stability, and vertical wind profiles that are often present in coastal regions during the summer (Thompson and Herman, 2019). Ozone lidar observations (Sullivan et al., 2014) (Figure 3d/3e, dashed black box) further illustrate the challenges of coastal AQ as it relates to the PBLH conceptual model in Figure 1. Both sites observe synoptic subsidence of long-range transport of increased ozone concentrations above 2 km AGL and have deep nocturnal residual layers that are entrained into the next day's PBL (sunrise is near 0600 LT and sunset is near 2030 LT on these days). Both ozonesondes and lidar profiles indicate the aloft ozone peak over HMI at ~1 km AGL is from the return flow of the bay breeze, which was also observed during the DISCOVER-AQ field campaign occurring in the same region (Loughner et al., 2014; Stauffer et al., 2015). Vertical stability was much greater over HMI, limiting the vertical mixing of the ozone pollution near the surface and the elevated layer at ~1 km. Lack of vertical mixing over water, or capping surface emissions in the lowest ~200 m AGL, indicates a critical need for understanding emissions over the water that are contributing to coastal AQ issues.

Figure 3 is representative of a real-world example illustrating the difficulties of quantifying land/water atmospheric gradients close to the surface. Accurate measurements from satellite instruments are challenging on this scale, if not impossible, due to the coarse satellite spatial resolution that results in mixed land-water pixels and significant land-water adjacency effects. Surface ozone (Figure 3a) is nearly 30 ppby greater over the water site (HMI) compared to the land-based locations during the ozonesonde launch, indicating a different photochemical regime than over land (where the surface values of ozone generally appear to be representative of the PBL estimate). However, these enhanced measurements make it possible to understand the pollution aloft more fully and evaluate transport patterns that would not have been measured in a traditional routine monitoring capacity. Ground-based networks of observing platforms, particularly during intensive campaign-based operations, are requisite to the understanding and evaluating the coastal PBLH and its exchange processes.

# 3. Representativeness of Satellite Measurements in Coastal Regions

Noting the ozone NAAQS non-attainment areas on Figure 2, we cannot currently measure PBL ozone from space at desired accuracy or precision from low Earth orbit satellites. However, commonly retrieved satellite quantities (below) serve as proxies for the spatial extent of coastal pollution. They illustrate successes and challenges for coastal measurements and are expected to benefit from upcoming geostationary satellite products.

#### 3.1. Satellite Measurements of PBLH

Estimates of the PBLH have been obtained directly from space and select detection methods are briefly summarized below for active sensing using elastic lidar and passive infrared sounding. In the literature there are many examples of estimates of PBLH from space-based systems that are used to better understand the seasonality of PBLH (e.g., McGrath-Spangler & Denning, 2012; Su et al., 2017) and evaluate regional and global model simulations of PBL (Hegarty et al., 2018; N. S. Jordan et al., 2010). GPS Radio Occultation (Ao et al., 2012; Kalmus et al., 2022) and passive infrared sounders, such as AIRS (Ding et al., 2021; J. P. A. Martins et al., 2010), are able to retrieve coarse PBLH estimates with horizontal ( $\sim 2^{\circ} \times 2^{\circ}$ -50 km × 50 km) and vertical (1–2 km) resolution based on atmospheric (e.g., temperature, moisture, or refractivity) gradients near the PBLH (Gettelman et al., 2004; Thrastarson et al., 2020) in coastal or other complex domains.

Under well-mixed (e.g., convective) PBL conditions, spaceborne lidars, such as the Cloud-Aerosol Transport System (CATS; Yorks et al., 2016), ICESat-2 (Markus et al., 2017), and CALIPSO (W. Zhang et al., 2016) are able to quantify daytime and nighttime PBLH with high vertical (30–100 m) precision using gradients in total (L1) or aerosol (L2) attenuated backscatter within a certain threshold as proxy for the PBLH. This retrieval is based on the assumption that the PBLH is located at the transition from the polluted PBL to the relatively less aerosol-laden air above the PBL However, in less idealized conditions (especially when stable stratification has developed below and above the thermodynamic PBLH), using attenuated backscatter presents a challenge to correctly interpret and algorithmically quantify (such as the wavelet covariance technique employed in Compton et al., 2013) an accurate PBLH. Furthermore, spaceborne lidars such as CALIPSO only visit within 10 km of

a specific ground pixel on average twice every 16 days (and one of these overpasses is in the nighttime hours), making the data set challenging to evaluate diurnal PBLH cycles in the coastal environment.

In summary, current spaceborne PBLH measurements provide either high-resolution vertical sampling with poor horizontal/temporal coverage, or near-once-daily global afternoon coverage with lower spatial resolution and accuracy. Although space-based datasets provide context for seasonality of the PBLH, which is important for global modeling and understanding long term climate trends, there are currently no spaceborne platforms that provide temporal or spatial PBL coverage beyond twice per day, much less at the horizontal spatial scales of a few km and vertical scales of ~30–100 m necessary for coastal areas. Furthermore, many of these platforms rely on the assumption of a strong gradient in a signal only at the top of the PBL and are likely unable to resolve mixed layering at multiple heights, especially layering that is above or within the PBL. Note that several groups are employing machine learning to identify where the PBL top is most likely located and to resolve the PBLH in areas of atmospheric stratification (Palm et al., 2021; Sleeman et al., 2020). Other investigations are using current and future low Earth orbit and geostationary passive sensors geared toward understanding the underlying atmospheric chemistry and dynamics of that impact PBLH retrievals in coastal regions (Loría-Salazar et al., 2021; Turner & Ulrich, 2021; Wang et al., 2021). Current research supported by NASA's PBL Decadal Survey Incubation program includes multi-sensor retrieval (passive and active) to take advantage of the combined capabilities and to integrate ground and spaceborne PBLH profiles into regional/global modeling data assimilation systems. There are challenges in validating novel retrievals; for example, high spatial and temporal resolution AQ models show sharp gradients in air pollution concentrations and nitrogen deposition near coastlines, but there is a lack of observations to evaluate these findings (Abdi-Oskouei et al., 2020; Loughner et al., 2016). Field data from US ozone NAAQS non-attainment areas (Figure 2), are still relatively sporadic, of limited duration (1-2 months); coastal PBL information is particularly lacking on deployments.

#### 3.2. Satellite Measurements of PBL Air Quality

Satellite measurements of tropospheric and total column amounts of nitrogen dioxide (NO<sub>2</sub>, Figure 2) have become increasingly accurate over the past two decades with improvements to retrieval algorithms from instruments like the GOME (Global Ozone Monitoring Experiment) series, Ozone Monitoring Instrument (OMI) and the higher spatial resolution TROPOspheric Monitoring Instrument (TROPOMI). However, relating the column amounts of pollutants to "nose-level" or PBL concentrations, especially in coastal regions, remains a challenge due to the complexities of vertical distribution and often strong spatial and temporal variability in surface (land or ocean) reflectivity (Knepp et al., 2015; Kollonige et al., 2018; Szykman et al., 2019; Thompson et al., 2019, 2023; Tzortziou et al., 2015, 2018). However, continued work is moving toward evaluating and utilizing these satellite derived NO<sub>2</sub> products to the neighborhood scale (Demetillo et al., 2021; Dressel et al., 2022; Goldberg et al., 2022).

Wet and dry atmospheric nitrogen deposition are major contributors of excess nitrogen (N) to coastal waters and ecosystems (Cornell et al., 2003; Kanakidou et al., 2016). Excessive reactive Nitrogen deposition (both wet and dry) can cause a series of negative effects on ecosystem health, biodiversity, soil, and water and has been summarized in a recent review by Liu et al., 2020. Methods have been established to estimate global NO<sub>2</sub> dry deposition fluxes at high spatial resolution using the combination of satellite measurements and high resolution chemical transport models (Geddes and Martin., 2017; Nowlan et al., 2014), but continue to be particularly challenging to correctly measure near coastal urban megacities or in instances of aged pollution transport (Kharol et al., 2018). In polluted coastal urban regions, excess N inputs from the atmosphere or land to adjacent aquatic systems may lead to eutrophication, recurrent algal blooms, and subsequent development of oxygen-depleted dead zones, such as those found almost every year in many heavily urbanized coasts and estuaries, for example, Gulf of Mexico, Chesapeake Bay, and Long Island Sound (Anderson & Taylor, 2001; Loughner et al., 2016; Murphy et al., 2011; Rabalais et al., 2001). Incorporated into vegetation directly through the leaves and indirectly through the soil, N deposition is also an important nutrient source for terrestrial plants and can affect biogeochemical and ecological processes across the continuum of terrestrial-aquatic ecosystems from upland forests to coastal wetlands (Morris, 1991).

#### 3.3. Satellite Measurements of Ocean Color

Remotely-sensed ocean color observations suggest that coastal PBL dynamics and AQ have profound effects on coastal water quality, aquatic ecosystems, and productivity. Over the past 20+ years, continuous measurements of

ocean color from polar orbiting satellites in low Earth orbit have quantified ocean biogeochemical cycles, ecological processes and their responses to climatic disturbances on a global scale (Werdell et al., 2018). Measurements are supplied by multi- and hyperspectral ocean color instruments, such as the NASA Sea-viewing Wide Field of View Sensor (SeaWiFS; 1997–2010), NASA's Moderate Resolution Imaging Spectroradiometers on Terra (MODIST; 1999–present) and Aqua (MODISA; 2002–present), ESA's Medium Resolution Imaging Spectrometer (MERIS; 2002–2012), the NASA-NOAA Suomi National Polar-orbiting Partnership Visible Infrared Imaging Radiometer Suite (VIIRS; 2012–present), ESA's Ocean and Land Color Instrument (OLCI; 2016–present), the Hyperspectral Imager for the Coastal Ocean (HICO, 2009–2014) and are expected from the upcoming Plankton, Aerosol, Cloud, ocean Ecosystem (PACE) mission. Measurements from regional geostationary sensors, such as the Geostationary Ocean Color Imager (GOCI mission, 2010–present, covering East Asia) provide unique information on highly dynamic physical processes and biogeochemical exchanges occurring in coastal waters. Note that reducing uncertainties for ocean properties through atmospheric correction of polar-orbiting and geostationary satellite data relies on knowing vertical and column amounts of aerosols and trace gases like ozone and NO<sub>2</sub> that may vary considerably across the land-water interface (Ahmad et al., 2007; Tzortziou et al., 2014).

A multi-satellite approach (for simultaneous retrievals of spatiotemporal variability in coastal atmospheric composition and ocean color) is thus required to obtain highest quality measurements of ocean biogeochemical properties. This fusion of multi-sensor data in turn will aid atmospheric retrievals because accurate aerosol and trace gas measurements are based on accurate knowledge of ocean surface reflectivity. Future measurements from upcoming ocean color hyperspectral (e.g., 2-nm to 5-nm spectral resolution) imagers, such as PACE and Geostationary Littoral Imaging and Monitoring Radiometer (GLIMR) may provide an opportunity to directly retrieve NO<sub>2</sub> over the ocean for both aquatic and atmospheric applications (Joiner et al., 2022). At present with only polar-orbiting satellites observing most of the globe (GOCI and GEMS over east Asia are exceptions) the inability to capture diurnal changes in atmospheric composition leads to uncertainties in non-coincident ocean color data. It is expected that being able to combine geostationary atmospheric and ocean color sensors TEMPO (Tropospheric Emissions: Monitoring of Pollution) and GLIMR over North America will provide an unprecedented opportunity to monitor the highly dynamic atmospheric and ocean processes along coastlines and to resolve important issues about PBL dynamics.

# 4. Summarizing Community Needs to Improve Coastal Air Quality Science

Recent coastal field campaign efforts listed in Table 1 have been funded and carried out in different ways depending on specific mission goals. However, several questions—of direct interest to coastal resource managers as well as scientists—remain in our understanding of the PBL and AQ in coastal environments. In summarizing the campaign efforts listed in Table 1 and satellite limitations, we recognize three areas of interdisciplinary research that are urgently needed to better understand the coupled PBL interactions.

- Improve fundamental understanding of the diurnal PBL cycle as it relates to recirculation and entrainment
  of nocturnal residual and long-range transport layers as they enter the coastal urban region. The campaigns
  broadly indicate that episodes where aloft layers or recirculation patterns persisted were the days generally
  driving exceedances of the ozone NAAQS. These features have occurred in several coastal regions (Table 1)
  as described for Long Island Sound (Rogers et al., 2020; Torres-Vazquez et al., 2022; Wu et al., 2021) and the
  Chesapeake Bay (Bernier et al., 2022; Dreessen et al., 2016; Kotsakis et al., 2022; Sullivan et al., 2017, 2019;
  Tao et al., 2022; Yang, Demoz, Delgado, Sullivan, et al., 2022, Yang, Demoz, Delgado, Tangborn, et al., 2022).
- 2. Quantify chemical perturbations from offshore transport of urban emissions, ships, boats (both commercial and personal/recreational) and offshore oil and gas drill sites, as they interact with the onset of a bay/sea breeze in the shallow marine boundary layer. Without accurate over/near water emissions, chemical transport models are not capable of simulating current and future regulatory scenarios (Sorte et al., 2020). Understanding these non-traditional emission sources will require the combination of satellite estimates and evaluations from sub-orbital campaign data sets. Preliminary studies reaching similar conclusions were documented in the Chesapeake Bay region (Caicedo et al., 2021; Gronoff et al., 2019; Ring et al., 2018), the South Korean coast waters (Thompson et al., 2019; Tzortziou et al., 2018), the Gulf of Mexico (Thompson et al., 2023), the Lake Michigan region (Vermeuel et al., 2019) and along the New York City coast (Nauth et al., 2023; Tzortziou et al., 2022).
- 3. Identify novel and viable pathways derived from the interaction of the land-ocean-atmosphere that can drive PBL content and structure in coastal communities. For example, describes aquatic vegetation contributing

to biogenic emissions that can produce pollution. Qin et al., 2019 demonstrates the importance of including a full inventory of biogenic emissions in modeling of coastal regions along the Great Lakes during LMOS (Table 1, 7). A recent review of how aquatic emissions can travel long distances to degrade AQ downstream appears in Lee et al. (2020).

The KORUS-OC/AQ campaign (see Crawford et al., 2021; C. E. Jordan et al., 2021a, 2021b; Thompson et al., 2019; Tzortziou et al., 2018) provides a model for future campaigns to examine both air and water quality issues at the coastal interface. Coincident intensive measurements of air pollution and water properties will advance our understanding of the feedbacks noted above that govern air/water exchange processes. Details on interdisciplinary topics needing more exploration are described in the NASA's Geostationary Coastal and Air Pollution Events report (GEOCAPE, https://geo-cape.larc.nasa.gov/wp-content/uploads/sites/142/2020/12/GEO-CAPE\_2018\_Final\_Report.pdf), with further details in the GEO-CAPE Interdisciplinary Science White Paper (https://geo-cape.larc.nasa.gov/wp-content/uploads/sites/142/2022/08/2022-08-29-Integrating-Coastal-Earth-System-Science-across-Marine\_19Jan.pdf).

The GEO-CAPE science team prioritized several interdisciplinary foci areas, with the most relevant being to form a better understanding of influences of meteorology on coastal waters and vice versa with the (a) role of coastal waters in land/sea breeze dynamics affecting transport and processing of atmospheric constituents and (b) changes in PBLH over estuarine and coastal waters with respect to that over their adjacent land masses and the subsequent effects on vertical distribution, transport, processing, and deposition of atmospheric constituents. Further examples of these that have not been explored or previously discussed include quantifying oil slicks in bodies of water and investigating the development and impacts of red tides and other algal blooms.

The 2017 Decadal Survey (NASEM, 2018) explicitly requested a satellite platform that could adequately resolve the large spatial variation of aerosol and trace gas concentrations (partly due to surface heterogeneities, such as urban-rural or land-water contrast), at a horizontal resolution of 5 km and a temporal resolution of every 2–3 hr. Therefore, the next generation of satellites needs to adopt new multi-sensor constellations with innovative deployment of small instruments or cubesats that can nimbly measure AQ in coastal environments along with ground-based systems used in prior field campaigns. Ground-based remote sensing and balloon-borne observations will continue to be vital to both evaluate the geo-stationary a priori models (Johnson et al., 2018) and to observe the fine spatial and temporal scales needed to understand coastal AQ within the PBL.

Launched over the next decade, geostationary sensors from the U.S. (TEMPO, GLIMR, and NOAA's Geostationary Extended Observations (GeoXO)), Europe (Sentinel 4), and SE Asia (GEMS and GOCI-II already in orbit) can form a northern hemisphere GEO constellation for studies of processes and exchanges across the land, ocean and atmosphere in complex coastal systems. Since these platforms will not have a dedicated PBL measurement onboard, the required approaches to improve our coastal PBL and AQ understanding remain: a dedicated sub-orbital airborne measurement suite to consistently resolve coastal PBL heights in a regional coastal domain where high resolution (<1 km) chemical transport model output can be confidently simulated. This is further enhanced with ground-based lidar/ceilometer, spectrometer/spectroradiometer, sun photometer, balloon, and in-situ measurements, particularly during intensive campaign-based operations. The combination of these suborbital observations (which are a fraction of the cost of any satellite program) can then continually validate existing satellite measurements and provide feedback to algorithm teams working on novel techniques, such as machine learning-based and multi-sensor (e.g., active + passive) PBL retrievals.

# 5. Conclusions

We described important land/water feedback processes in the PBL, discussed measurement strategies to better understand these processes, and identified critical measurement gaps of PBL characteristics that must be addressed to fully characterize the interaction of AQ, physical and biological processes along the coast. For the first time, we have summarized scientific consensus from analyses during coastal AQ campaigns (Table 1) and these results have been used to highlight the complexities of real-world PBL measurements over the water (Figure 3).

Continental air pollution emission sources are generally well-known (e.g., transportation, industry, and biogenic) but characterizing their downstream effects on water quality where PBL dynamics are complex remains a challenge. Most coastal areas observe large differences in model projections of air pollution compared to observations, which prevents policymakers from designing sound implementation plans for improving coastal air and



water quality. A community driven design for future generations of spaceborne measurements, model simulations, and Earth System science will produce effective regulatory mitigation efforts, enhance environmental justice, and improve public health. However, this will only be accomplished with the combination of interdisciplinary research support and consistent/routine suborbital observations.

#### **Data Availability Statement**

OMI version 4 tropospheric NO<sub>2</sub> averages can be downloaded from the NASA Goddard Space Flight Center Aura Validation Data Center (Krotkov et al., 2019). Data for US EPA Non-Attainment Areas for criteria pollutants can be accessed at https://www.epa.gov/green-book/green-book-gis-download (US Environmental Protection Agency, 2023). The tropospheric ozone lidar used from NASA's Tropospheric Ozone Lidar Network (TOLNet) and are publicly available (https://www-air.larc.nasa.gov/cgi-bin/ArcView.1/TOLNet?NASA-GSFC=1, Newchurch et al., 2016). The ozonesonde data used in this publication were obtained as part of the OWLETS-2 campaign and are publicly available (https://www-air.larc.nasa.gov/missions/owlets, following procedures documented in Thompson et al., 2019).

# References

- Abdi-Oskouei, M., Carmichael, G., Christiansen, M., Ferrada, G., Roozitalab, B., Sobhani, N., et al. (2020). Sensitivity of meteorological skill to selection of WRF-Chem physical parameterizations and impact on ozone prediction during the Lake Michigan Ozone Study (LMOS). Journal of Geophysical Research: Atmospheres, 125(5), e2019JD031971. https://doi.org/10.1029/2019jd031971
- Ahmad, Z., McClain, C. R., Herman, J. R., Franz, B., Kwiatkowska, E., Robinson, W., et al. (2007). Atmospheric correction for NO<sub>2</sub> absorption in retrieving water-leaving reflectances from the SeaWiFS and MODIS measurements. *Applied Optics*, 46(26), 6504–6512. https://doi.org/10.1364/AO.46.006504
- Anderson, T. H., & Taylor, G. T. (2001). Nutrient pulses, plankton blooms, and seasonal hypoxia in western Long Island Sound. *Estuaries*, 24(2), 228–243. https://doi.org/10.2307/1352947
- Anenberg, S. C., Mohegh, A., Goldberg, D. L., Kerr, G. H., Brauer, M., Burkart, K., et al. (2022). Long-term trends in urban NO<sub>2</sub> concentrations and associated paediatric asthma incidence: Estimates from global datasets. *The Lancet Planetary Health*, 6(1), e49–e58. https://doi. org/10.1016/s2542-5196(21)00255-2
- Ao, C. O., Waliser, D. E., Chan, S. K., Li, J.-L., Tian, B., Xie, F., & Manucci, A. J. (2012). Planetary boundary layer heights from GPS radio occultation refractivity and humidity profiles. *Journal of Geophysical Research*, 117(D16). https://doi.org/10.1029/2012JD017598
- Behrenfeld, M. J., Moore, R. H., Hostetler, C. A., Graff, J., Gaube, P., Russell, L. M., et al. (2019). The North Atlantic aerosol and marine ecosystem study (NAAMES): Science motive and mission overview. Frontiers in Marine Science, 6, 122. https://doi.org/10.3389/fmars.2019.00122
- Bernier, C., Wang, Y., Gronoff, G., Berkoff, T., Knowland, K. E., Sullivan, J., et al. (2022). Cluster-based characterization of multi-dimensional tropospheric ozone variability in coastal regions: An analysis of lidar measurements and model results. *Atmospheric Chemistry and Physics Discussions*, 1–27(23), 15313–15331. https://doi.org/10.5194/acp-22-15313-2022
- Brown, S., Nicholls, R., Woodroffe, C., Hanson, S., Hinkel, J., Kebede, A. S., et al. (2013). Sea-level rise impacts and responses: A global perspective. In C. W. Finkl (Ed.), *Coastal hazards* (pp. 117–149). Springer.
- Caicedo, V., Delgado, R., Luke, W. T., Ren, X., Kelley, P., Stratton, P. R., et al. (2021). Observations of bay-breeze and ozone events over a marine site during the OWLETS-2 campaign. Atmospheric Environment, 263, 118669. https://doi.org/10.1016/j.atmosenv.2021.118669
- Clune, J. W., Capel, P. D., Miller, M. P., Burns, D. A., Sekellick, A. J., Claggett, P. R., et al. (2021). Nitrogen in the Chesapeake Bay watershed—A century of change, 1950–2050. No. 1486. US Geological Survey.
- Compton, J. C., Delgado, R., Berkoff, T. A., & Hoff, R. M. (2013). Determination of planetary boundary layer height on short spatial and temporal scales: A demonstration of the covariance wavelet transform in ground-based wind profiler and lidar measurements. *Journal of Atmospheric* and Oceanic Technology, 30(7), 1566–1575. https://doi.org/10.1175/jtech-d-12-00116.1
- Cornell, S. E., Jickells, T. D., Cape, J. N., Rowland, A. P., & Duce, R. A. (2003). Organic nitrogen deposition on land and coastal environments: A review of methods and data. *Atmospheric Environment*, 37(16), 2173–2191. https://doi.org/10.1016/s1352-2310(03)00133-x
- Crawford, J. H., Ahn, J. Y., Al-Saadi, J., Chang, L., Emmons, L. K., Kim, J., et al. (2021). The Korea–United States Air Quality (KORUS-AQ) field study. *Elem Sci Anth*, 9(1), 00163. https://doi.org/10.1525/elementa.2020.00163
- Crosman, E. T., & Horel, J. D. (2010). Sea and lake breezes: A review of numerical studies. *Boundary-Layer Meteorology*, 137(1), 1–29. https://doi.org/10.1007/s10546-010-9517-9
- Crossett, K., Culliton, T., Wiley, P., & Goodspeed, T. (2004). Population trends along the coastal United States, 1980–2008 (47p.). Silver Spring, National Oceanic and Atmospheric Administration.
- Dacic, N., Sullivan, J. T., Knowland, K. E., Wolfe, G. M., Oman, L. D., Berkoff, T. A., & Gronoff, G. P. (2020). Evaluation of NASA's high-resolution global composition simulations: Understanding a pollution event in the Chesapeake Bay during the summer 2017 OWLETS campaign. Atmospheric Environment, 222, 117133. https://doi.org/10.1016/j.atmosenv.2019.117133
- Demetillo, M. A. G., Harkins, C., McDonald, B. C., Chodrow, P. S., Sun, K., & Pusede, S. R. (2021). Space-based observational constraints on NO<sub>2</sub> air pollution inequality from diesel traffic in major US cities. *Geophysical Research Letters*, 48(17), e2021GL094333. https://doi. org/10.1029/2021gl094333
- Ding, F., Iredell, L., Theobald, M., Wei, J., & Meyer, D. (2021). PBL height from AIRS, GPS RO, and MERRA-2 products in NASA GES DISC and their 10-year seasonal mean intercomparison. *Earth and Space Science*, 8(9), e2021EA001859. https://doi.org/10.1029/2021EA001859
- Dreessen, J., Ren, X., Gardner, D., Green, K., Stratton, P., Sullivan, J. T., et al. (2023). VOC and trace gas measurements and ozone chemistry over the Chesapeake Bay during OWLETS-2, 2013. *Journal of the Air & Waste Management Association*, 73(3), 178–199. https://doi.org/10. 1080/10962247.2022.2136782
- Dreessen, J., Sullivan, J., & Delgado, R. (2016). Observations and impacts of transported Canadian wildfire smoke on ozone and aerosol air quality in the Maryland region on June 9–12, 2015. *Journal of the Air & Waste Management Association*, 66(9), 842–862. https://doi.org/10. 1080/10962247.2016.1161674

#### Acknowledgments

This work was supported by the NASA HQ Tropospheric Composition, Upper Atmospheric Research Program and NASA Earth Science Directorate Decadal Survey (with a special thanks to Robert J. Swap for further support and guidance). The authors graciously thank our university, federal, and local collaborators (especially Alexander E. Kotsakis and Nader Abuhassan) that have made up the entirety of the various campaign teams who provided insight on these discussions and personal communications. We especially thank those individuals involved in the ozonesonde launches and coordination used in the figures in this work: Laurence W. Twigg, Adrian Flores, Ricardo Sakai, Ruben Delgado, Timothy Berkoff, Joel Dreessen, and Tad Aburn.

- Dressel, I. M., Demetillo, M. A. G., Judd, L. M., Janz, S. J., Fields, K. P., Sun, K., et al. (2022). Daily satellite observations of nitrogen dioxide air pollution inequality in New York City, New York and Newark, New Jersey: Evaluation and application. *Environmental Science & Technology*, 56(22), 15298–15311. https://doi.org/10.1021/acs.est.2c02828
- Geddes, J. A., & Martin, R. V. (2017). Global deposition of total reactive nitrogen oxides from 1996 to 2014 constrained with satellite observations of NO, columns. Atmospheric Chemistry and Physics, 17(16), 10071–10091. https://doi.org/10.5194/acp-17-10071-2017
- Gettelman, A., Weinstock, E. M., Fetzer, E. J., Irion, F. W., Eldering, A., Richard, E. C., et al. (2004). Validation of Aqua satellite data in the upper troposphere and lower stratosphere with in situ aircraft instruments. *Geophysical Research Letters*, 31(22). https://doi.org/10.1029/2004gl020730
- Goldberg, D. L., Anenberg, S. C., Kerr, G. H., Mohegh, A., Lu, Z., & Streets, D. G. (2021). TROPOMI NO<sub>2</sub> in the United States: A detailed look at the annual averages, weekly cycles, effects of temperature, and correlation with surface NO<sub>2</sub> concentrations. *Earth's Future*, 9(4), e2020EF001665. https://doi.org/10.1029/2020ef001665
- Goldberg, D. L., Loughner, C. P., Tzortziou, M., Stehr, J. W., Pickering, K. E., Marufu, L. T., & Dickerson, R. R. (2014). Higher surface ozone concentrations over the Chesapeake Bay than over the adjacent land: Observations and models from the DISCOVER-AQ and CBODAQ campaigns. Atmospheric Environment, 84, 9–19. https://doi.org/10.1016/j.atmosenv.2013.11.008
- Gronoff, G., Robinson, J., Berkoff, T., Swap, R., Farris, B., Schroeder, J., et al. (2019). A method for quantifying near range point source induced O<sub>3</sub> titration events using Co-located Lidar and Pandora measurements. *Atmospheric Environment*, 204, 43–52. https://doi.org/10.1016/j. atmosenv.2019.01.052
- Hegarty, J. D., Lewis, J., McGrath-Spangler, E. L., Henderson, J., Scarino, A. J., DeCola, P., et al. (2018). Analysis of the planetary boundary layer height during DISCOVER-AQ Baltimore–Washington, DC, with lidar and high-resolution WRF modeling. *Journal of Applied Meteorology* and Climatology, 57(11), 2679–2696. https://doi.org/10.1175/jamc-d-18-0014.1
- Jensen, M. P., Flynn, J. H., Judd, L. M., Kollias, P., Kuang, C., Mcfarquhar, G., et al. (2022). A succession of cloud, precipitation, aerosol, and air quality field experiments in the coastal urban environment. *Bulletin of the American Meteorological Society*, 103(2), 103–105. https://doi. org/10.1175/bams-d-21-0104.1
- Johnson, M. S., Liu, X., Zoogman, P., Sullivan, J., Newchurch, M., Kuang, S., et al. (2018). Evaluation of potential sources of a priori ozone profiles for TEMPO tropospheric ozone retrievals. Atmospheric Measurement Techniques, 11(6), 3457–3477. https://doi.org/10.5194/ amt-11-3457-2018
- Johnson, M. S., Philip, S., Kumar, R., Naeger, A., Souri, A. H., Geddes, J., et al. (2022). Satellite remote-sensing capability to assess tropospheric column ratios of formaldehyde and nitrogen dioxide: Case study during the LISTOS 2018 field campaign. Atmospheric Measurement Techniques Discussions, 1–41.
- Joiner, J., Marchenko, S., Fasnacht, Z., Lamsal, L., Li, C., Vasilkov, A., & Krotkov, N. (2022). Use of machine learning to retrieve nitrogen dioxide with hyperspectral imagers in the ultraviolet and blue spectral range (pp. 1–27). EGUsphere.
- Jordan, C. E., Stauffer, R. M., Lamb, B. T., Hudgins, C. H., Thornhill, K. L., Schuster, G. L., et al. (2021a). New in situ aerosol hyperspectral optical measurements over 300–700 nm – Part 1: Spectral Aerosol Extinction (SpEx) instrument field validation during the KORUS-OC cruise. Atmospheric Measurement Techniques, 14(1), 695–713. https://doi.org/10.5194/amt-14-695-2021
- Jordan, C. E., Stauffer, R. M., Lamb, B. T., Novak, M., Mannino, A., Crosbie, E. C., et al. (2021b). New in situ aerosol hyperspectral optical measurements over 300–700 nm – Part 2: Extinction, total absorption, water-and methanol-soluble absorption observed during the KORUS-OC cruise. Atmospheric Measurement Techniques, 14(1), 715–736. https://doi.org/10.5194/amt-14-715-2021
- Jordan, N. S., Hoff, R. M., & Bacmeister, J. T. (2010). Validation of Goddard Earth Observing System-version 5 MERRA planetary boundary layer heights using CALIPSO. Journal of Geophysical Research, 115(D24). https://doi.org/10.1029/2009JD013777
- Judd, L. M., Al-Saadi, J. A., Szykman, J. J., Valin, L. C., Janz, S. J., Kowalewski, M. G., et al. (2020). Evaluating Sentinel-5P TROPOMI tropospheric NO 2 column densities with airborne and Pandora spectrometers near New York City and Long Island Sound. Atmospheric Measurement Techniques, 13(11), 6113–6140. https://doi.org/10.5194/antt-13-6113-2020
- Judd, L. M., Sullivan, J. T., Lefer, B., Haynes, J., Jensen, M. P., & Nadkarni, R. (2021). TRACER-AQ science plan: An interagency cooperative air quality field study in the Houston, TX Metropolitan Region. Retrieved from https://www-air.larc.nasa.gov/missions/tracer-aq/docs/ TRACERAQ\_SciencePlan\_v1.pdf
- Kalmus, P., Ao, C. O., Wang, K.-N., Manzi, M. P., & Teixiera, J. (2022). A high-resolution planetary boundary layer height seasonal climatology from GNSS radio occultations. *Remote Sensing of Environment*, 276, 113037. https://doi.org/10.1016/j.rse.2022.113037
- Kanakidou, M., Myriokefalitakis, S., Daskalakis, N., Fanourgakis, G., Nenes, A., Baker, A. R., et al. (2016). Past, present, and future atmospheric nitrogen deposition. *Journal of the Atmospheric Sciences*, 73(5), 2039–2047. https://doi.org/10.1175/jas-d-15-0278.1
- Karambelas, A. (2020). LISTOS: Toward a better understanding of New York City's ozone pollution problem. EM.
- Kharol, S. K., Shephard, M. W., McLinden, C. A., Zhang, L., Sioris, C. E., O'Brien, J. M., et al. (2018). Dry deposition of reactive nitrogen from satellite observations of ammonia and nitrogen dioxide over North America. *Geophysical Research Letters*, 45(2), 1157–1166. https://doi. org/10.1002/2017gl075832
- Knepp, T., Pippin, M., Crawford, J., Chen, G., Szykman, J., Long, R., et al. (2015). Estimating surface NO<sub>2</sub> and SO<sub>2</sub> mixing ratios from fast response total column observations and potential application to geostationary missions. *Journal of Atmospheric Chemistry*, 72(3–4), 261–286. https://doi.org/10.1007/s10874-013-9257-6
- Kollonige, D. E., Thompson, A. M., Josipovic, M., Tzortziou, M., Beukes, J. P., Burger, R., et al. (2018). OMI satellite and ground-based Pandora observations and their application to surface NO<sub>2</sub> estimations at terrestrial and marine sites. *Journal of Geophysical Research: Atmospheres*, 123(2), 1441–1459. https://doi.org/10.1002/2017JD026518
- Kotsakis, A., Sullivan, J. T., Hanisco, T. F., Swap, R. J., Caicedo, V., Berkoff, T. A., et al. (2022). Sensitivity of total column NO<sub>2</sub> at a marine site within the Chesapeake Bay during OWLETS-2. Atmospheric Environment, 277, 119063. https://doi.org/10.1016/j.atmosenv.2022.119063
- Kowalewski, M. J., & Janz, S. J. (2014). Remote sensing capabilities of the GeoCAPE Airborne Simulator. In Earth observing systems XIX, 921811. Proceedings of SPIE (Vol. 9218). https://doi.org/10.1117/12.2062058
- Krotkov, N. A., Lamsal, L. N., Marchenko, S. V., Bucsela, E. J., Swartz, W. H., Joiner, J., et al. (2019). OMI/aura nitrogen dioxide (NO<sub>2</sub>) total and tropospheric column 1-orbit L2 Swath 13 × 24 km V003. Goddard Earth Sciences Data and Information Services Center (GES DISC). https:// doi.org/10.5067/Aura/OMI/DATA2017
- Lee, B., Yi, X., Hadibarata, T., & Yuniarto, A. (2020). Phytoremediation mechanisms in air pollution control: A review. Water, Air, & Soil Pollution, 231(8), 1–13. https://doi.org/10.1007/s11270-020-04813-6
- Liu, L., Zhang, X., Xu, W., Liu, X., Lu, X., Wei, J., et al. (2020). Reviewing global estimates of surface reactive nitrogen concentration and deposition using satellite retrievals. Atmospheric Chemistry and Physics, 20(14), 8641–8658. https://doi.org/10.5194/acp-20-8641-2020
- Loría-Salazar, S. M., Sayer, A. M., Barnes, J., Huang, J., Flynn, C., Lareau, N., et al. (2021). Evaluation of novel NASA MODerate Resolution Imaging Spectroradiometer and Visible Infrared Imaging Radiometer Suite aerosol products and assessment of smoke height boundary layer

ratio during extreme smoke events in the western USA. Journal of Geophysical Research: Atmospheres, 126(11), e2020JD034180. https://doi.org/10.1029/2020jd034180

- Loughner, C. P., Allen, D. J., Pickering, K. E., Zhang, D.-L., Shou, Y.-X., & Dickerson, R. R. (2011). Impact of fair-weather cumulus clouds and the Chesapeake Bay breeze on pollutant transport and transformation. *Atmospheric Environment*, 45(24), 4060–4072. https://doi.org/10.1016/j. atmosenv.2011.04.003
- Loughner, C. P., Tzortziou, M., Follette-Cook, M., Pickering, K. E., Goldberg, D., Satam, C., et al. (2014). Impact of bay-breeze circulations on surface air quality and boundary layer export. *Journal of Applied Meteorology and Climatology*, 53(7), 1697–1713. https://doi.org/10.1175/ JAMC-D-13-0323.1
- Loughner, C. P., Tzortziou, M., Shroder, S., & Pickering, K. E. (2016). Enhanced dry deposition of nitrogen pollution near coastlines: A case study covering the Chesapeake Bay estuary and Atlantic Ocean coastline. *Journal of Geophysical Research: Atmospheres*, 121(23), 14221– 14238. https://doi.org/10.1002/2016JD025571
- McGrath-Spangler, E. L., & Denning, A. S. (2012). Estimates of North American summertime planetary boundary layer depths derived from space-borne lidar. Journal of Geophysical Research, 117(D15). https://doi.org/10.1029/2012JD017615
- Markus, T., Neumann, T., Martino, A., Abdalati, W., Brunt, K., Csatho, B., et al. (2017). The Ice, Cloud, and land Elevation Satellite-2 (ICESat-2): Science requirements, concept, and implementation. *Remote Sensing of Environment*, 190, 260–273. https://doi.org/10.1016/j.rse.2016.12.029
- Martins, D. K., Najjar, R. G., Tzortziou, M., Abuhassan, N., Thompson, A. M., & Kollonige, D. E. (2016). Spatial and temporal variability of ground and satellite column measurements of NO<sub>2</sub> and O<sub>3</sub> over the Atlantic ocean during the Deposition of Atmospheric Nitrogen to Coastal Ecosystems Experiment (DANCE). Journal of Geophysical Research: Atmospheres, 121(23), 175–114. https://doi.org/10.1002/2016JD024998
- Martins, J. P. A., Teixeira, J., Soares, P. M. M., Miranda, P. M. A., Kahn, B. H., Dang, V. T., et al. (2010). Infrared sounding of the trade-wind boundary layer: AIRS and the RICO experiment. *Geophysical Research Letters*, 37(24), L24806. https://doi.org/10.1029/2010GL045902
- Mazzuca, G. M., Pickering, K. E., New, D. A., Dreessen, J., & Dickerson, R. R. (2019). Impact of bay breeze and thunderstorm circulations on surface ozone at a site along the Chesapeake Bay 2011–2016. Atmospheric Environment, 198, 351–365. https://doi.org/10.1016/j. atmosenv.2018.10.068
- Medeiros, B., Hall, A., & Stevens, B. (2005). What controls the mean depth of the PBL? Journal of Climate, 18(16), 3157–3172. https://doi.org/10.1175/jcli3417.1
- Miller, S. T. K., Keim, B. D., Talbot, R. W., & Huiting Mao, H. (2003). Sea breeze: Structure, forecasting, and impacts. *Reviews of Geophysics*, 41(3). https://doi.org/10.1029/2003RG000124
- Morris, J. T. (1991). Effects of nitrogen loading on wetland ecosystems with particular reference to atmospheric deposition. Annual Review of Ecology and Systematics, 22(1), 257–279. https://doi.org/10.1146/annurev.es.22.110191.001353
- Murphy, R. R., Kemp, W. M., & Ball, W. P. (2011). Long-term trends in Chesapeake Bay seasonal hypoxia, stratification, and nutrient loading. *Estuaries and Coasts*, 34(6), 1293–1309. https://doi.org/10.1007/s12237-011-9413-7
- National Academies of Sciences, Engineering, and Medicine. (2018). Thriving on our changing planet: A decadal strategy for Earth observation from space. The National Academies Press. https://doi.org/10.17226/24938
- Nauth, D. (2021). Analyzing Atmospheric total column NO<sub>2</sub> (TCNO<sub>2</sub>) and surface NO<sub>2</sub> variability in the New York City area using ground- and satellite-based instruments. MSc thesis dissertation, The City College of New York.
- Nauth, D., Loughner, C. P., & Tzortziou, M. (2023). The influence of synoptic scale wind patterns on column integrated nitrogen dioxide, ground level ozone, and the development of sea breeze circulations in the New York City metropolitan area. *Journal of Applied Meteorology and Climatology*. https://doi.org/10.1175/jamc-d-22-0145.1
- Newchurch, M. J., Kuang, S., Leblanc, T., Alvarez, R. J., Langford, A. O., Senff, C. J., et al. (2016). TOLNET–A tropospheric ozone lidar profiling network for satellite continuity and process studies. In EPJ web of conferences (Vol. 119, p. 20001). EDP Sciences.
- Nowlan, C. R., Martin, R. V., PhilipLamsal, S. L. N., Krotkov, N. A., Marais, E. A., & Zhang, Q. (2014). Global dry deposition of nitrogen dioxide and sulfur dioxide inferred from space-based measurements. *Global Biogeochemical Cycles*, 28(10), 1025–1043. https://doi. org/10.1002/2014gb004805
- OMI. OMI version 4 tropospheric NO<sub>2</sub> averages can be downloaded from the NASA Goddard Space Flight Center aura Validation Data Center. Retrieved from https://avdc.gsfc.nasa.gov/pub/data/satellite/Aura/OMI/V03/L3/
- Palm, S. P., Selmer, P., Yorks, J., Nicholls, S., & Nowottnick, E. (2021). Planetary boundary layer height estimates from ICESat-2 and CATS backscatter measurements. *Frontiers in Remote Sensing*, 2, 716951. https://doi.org/10.3389/frsen.2021.716951
- Peterson, D. A., Hyer, E. J., Han, S.-O., Crawford, J. H., Park, R. J., Holz, R., et al. (2019). Meteorology influencing springtime air quality, pollution transport, and visibility in Korea. *Elementa: Science of the Anthropocene*, 7, 57. https://doi.org/10.1525/elementa.395
- Qin, M., Yu, H., Hu, Y., Russell, A. G., Odman, M. T., Doty, K., et al. (2019). Improving ozone simulations in the Great Lakes Region: The role of emissions, chemistry, and dry deposition. *Atmospheric Environment*, 202, 167–179. https://doi.org/10.1016/j.atmosenv.2019.01.025
- Rabalais, N. N., Turner, R. E., & Wiseman, W. J., Jr. (2001). Hypoxia in the Gulf of Mexico. *Journal of Environmental Quality*, 30(2), 320–329. https://doi.org/10.2134/jeq2001.302320x
   Ring, A. M. Canty, T. P. Anderson, D. C. Vinciguerra, T. P. He, H. Goldberg, D. L. et al. (2018). Evaluating commercial marine.
- Ring, A. M., Canty, T. P., Anderson, D. C., Vinciguerra, T. P., He, H., Goldberg, D. L., et al. (2018). Evaluating commercial marine emissions and their role in air quality policy using observations and the CMAQ model. *Atmospheric Environment*, 173, 96–107. https://doi.org/10.1016/j.atmosenv.2017.10.037
- Rogers, H. M., Ditto, J. C., & Gentner, D. R. (2020). Evidence for impacts on surface-level air quality in the northeastern US from long-distance transport of smoke from North American fires during the Long Island Sound Tropospheric Ozone Study (LISTOS) 2018. Atmospheric Chemistry and Physics, 20(2), 671–682. https://doi.org/10.5194/acp-20-671-202
- Santanello, J. A., Dirmeyer, P. A., Ferguson, C. R., Findell, K. L., Tawfik, A. B., Berg, A., et al. (2018). Land-atmosphere interactions: The LoCo Perspective. Bulletin of the American Meteorological Society, 99(6), 1253–1272. https://doi.org/10.1175/BAMS-D-17-0001.1
- Sleeman, J., Halem, M., Yang, Z., Caicedo, V., Demoz, B., & Delgado, R. (2020). A deep machine learning approach for lidar based boundary layer height detection. In *IGARSS 2020-2020 IEEE international geoscience and remote sensing symposium* (pp. 3676–3679). IEEE.
- Sorte, S., Rodrigues, V., Borrego, C., & Monteiro, A. (2020). Impact of harbour activities on local air quality: A review. *Environmental Pollution*, 257, 113542. https://doi.org/10.1016/j.envpol.2019.113542
- Stanier, C. O., Pierce, R. B., Abdi-Oskouei, M., Adelman, Z. E., Al-Saadi, J., Alwe, H. D., et al. (2021). Overview of the Lake Michigan ozone study 2017. Bulletin of the American Meteorological Society, 102(12), E2207–E2225. https://doi.org/10.1175/BAMS-D-20-0061.1
- Stauffer, R. M., & Thompson, A. M. (2015). Bay breeze climatology at two sites along the Chesapeake Bay from 1986–2010: Implications for surface ozone. *Journal of Atmospheric Chemistry*, 72(3–4), 355–372. https://doi.org/10.1007/s10874-013-9260-y
- Stauffer, R. M., Thompson, A. M., Martins, D. K., Clark, R. D., Goldberg, D. L., Loughner, C. P., et al. (2015). Bay breeze influence on surface ozone at Edgewood, MD during July 2011. Journal of Atmospheric Chemistry, 72(3–4), 335–353. https://doi.org/10.1007/s10874-012-9241-6



- Su, T., Li, J., Li, C., Xiang, P., Lau, A. K.-H., Guo, J., et al. (2017). An intercomparison of long-term planetary boundary layer heights retrieved from CALIPSO, ground-based lidar, and radiosonde measurements over Hong Kong. *Journal of Geophysical Research: Atmospheres*, 122(7), 3929–3943. https://doi.org/10.1002/2016JD025937
- Sullivan, J. T., Berkoff, T., Gronoff, G., Knepp, T., Pippin, M., Allen, D., et al. (2019). The Ozone Water–Land Environmental Transition Study: An innovative strategy for understanding Chesapeake Bay pollution events. *Bulletin of the American Meteorological Society*, 100(2), 291–306. https://doi.org/10.1175/BAMS-D-18-0025.1
- Sullivan, J. T., McGee, T. J., Sumnicht, G. K., Twigg, L. W., & Hoff, R. M. (2014). A mobile differential absorption lidar to measure sub-hourly fluctuation of tropospheric ozone profiles in the Baltimore–Washington, DC region. *Atmospheric Measurement Techniques*, 7(10), 3529– 3548. https://doi.org/10.5194/amt-7-3529-2014
- Sullivan, J. T., Rabenhorst, S. D., Dreessen, J., McGee, T. J., Delgado, R., Twigg, L., & Sumnicht, G. (2017). Lidar observations revealing transport of O<sub>3</sub> in the presence of a nocturnal low-level jet: Regional implications for "next-day" pollution. *Atmospheric Environment*, 158, 160–171. https://doi.org/10.1016/j.atmosenv.2017.03.039
- Szykman, J., Swap, R. J., Lefer, B., Valin, L., Lee, C., Fioletov, V., et al. (2019). Pandora connecting in-situ and satellite monitoring in support of the Canada–U.S. Air Quality Agreement. The Magazine for Environmental Managers A&WMA.
- Tao, M., Fiore, A. M., Jin, X., Schiferl, L. D., Commane, R., Judd, L. M., et al. (2022). Investigating changes in ozone formation chemistry during summertime pollution events over the Northeastern United States. *Environmental Science & Technology*, 56(22), 15312–15327. https://doi. org/10.1021/acs.est.2c02972
- Teixeira, J., Piepmeier, J. R., Nehrir, A. R., Ao, C. O., Chen, S. S., Clayson, C. A., et al. (2021). Toward a global planetary boundary layer observing system: The NASA PBL incubation study team report (Vol. 134). NASA PBL Incubation Study Team. Retrieved from https://science.nasa. gov/science-pink/s3fs-public/atoms/files/NASAPBLIncubationFinalReport.pdf
- Thompson, A. M., Kollonige, D. E., Stauffer, R. M., Abuhassan, N., Kotsakis, A. E., Lamsal, L. N., et al. (2023). Two air quality regimes in total column NO<sub>2</sub> over the Gulf of Mexico in May 2019: Shipboard and satellite views. *Earth and Space Science*, *10*(3). https://doi. org/10.1029/2022EA002473
- Thompson, A. M., Kollonige, D. E., Stauffer, R. M., Abuhassan, N., Kotsakis, A. E., Swap, R. J., & Wecht, H. (2020). SCOAPE: Satellite and shipboard views of Air quality along the Louisiana coast. EM.
- Thompson, A. M., Smit, H. G. J., Witte, J. C., Stauffer, R. M., Johnson, B. J., Morris, G., et al. (2019). Ozonesonde quality assurance: The JOSIE– SHADOZ (2017) experience. Bulletin of the American Meteorological Society, 100(1), 155–171. https://doi.org/10.1175/bams-d-17-0311.1
- Thompson, A. M., Stauffer, R. M., Boyle, T. P., Kollonige, D. E., Miyazaki, K., Tzortziou, M., et al. (2019). Comparison of near-surface NO<sub>2</sub> pollution with Pandora total column NO<sub>2</sub> during the Korea-United States Ocean Color (KORUS OC) campaign. *Journal of Geophysical Research: Atmospheres*, 124(23), 13560–13575. https://doi.org/10.1029/2019JD030765
- Thrastarson, H. T., Manning, E., Kahn, B., Fetzer, E., Yue, Q., Wong, S., et al. (2020). AIRS/AMSU/HSB version 7 Level 2 product user guide (pp. 83–92). Jet Propulsion Laboratory, California Institute of Technology. Retrieved from https://docserver.gesdisc.eosdis.nasa.gov/public/ project/AIRS/V7\_L2\_Product\_User\_Guide.pdf
- Torres-Vazquez, A., Pleim, J., Gilliam, R., & Pouliot, G. (2022). Performance evaluation of the meteorology and air quality conditions from multiscale WRF-CMAQ simulations for the Long Island Sound Tropospheric Ozone Study (LISTOS). Journal of Geophysical Research: Atmospheres, 127(5), e2021JD035890. https://doi.org/10.1029/2021jd035890
- Turner, D. D., & Ulrich, L. (2021). Ground-based temperature and humidity profiling: Combining active and passive remote sensors. Atmospheric Measurement Techniques, 14(4), 3033–3048. https://doi.org/10.5194/amt-14-3033-2021
- Tzortziou, M., Herman, J. R., Ahmad, Z., Loughner, C. P., Abuhassan, N., & Cede, A. (2014). Atmospheric NO<sub>2</sub> dynamics and impact on ocean color retrievals in urban nearshore regions. *Journal of Geophysical Research-Oceans*, 119(6), 3834–3854. https://doi.org/10.1002/2014JC009803
- Tzortziou, M., Herman, J. R., Cede, A., Loughner, C. P., Abuhassan, N., & Naik, S. (2015). Spatial and temporal variability of ozone and nitrogen dioxide over a major urban estuarine ecosystem. *Journal of Atmospheric Chemistry*, 72(3–4), 287–309. https://doi.org/10.1007/ s10874-013-9255-8
- Tzortziou, M., Kwong, C. F., Goldberg, D., Schiferl, L., Commane, R., Abuhassan, N., et al. (2022). Declines and peaks in NO<sub>2</sub> pollution during the multiple waves of the COVID-19 pandemic in the New York metropolitan area. *Atmospheric Chemistry and Physics*, 22(4), 2399–2417. https://doi.org/10.5194/acp-22-2399-2022
- Tzortziou, M., Parker, O., Lamb, B., Herman, J. R., Lamsal, L., Stauffer, R., & Abuhassan, N. (2018). Atmospheric trace gas (NO<sub>2</sub> and O<sub>3</sub>) variability in South Korean coastal waters, and implications for remote sensing of coastal ocean color dynamics. *Remote Sensing*, *10*, 1587. https://doi.org/10.3390/rs10101587
- US Environmental Protection Agency. (2023). The green book nonattainment areas for criteria pollutants.
- Vermeuel, M. P., Novak, G. A., Alwe, H. D., Hughes, D. D., Kaleel, R., Dickens, A. F., et al. (2019). Sensitivity of ozone production to NOx and VOC along the Lake Michigan coastline. *Journal of Geophysical Research: Atmospheres*, 124(20), 10989–11006. https://doi. org/10.1029/2019JD030842
- Wang, Y., Wang, J., Levy, R. C., Shi, Y. R., Mattoo, S., & Reid, J. S. (2021). First retrieval of AOD at fine resolution over shallow and turbid coastal waters from MODIS. *Geophysical Research Letters*, 48(17), e2021GL094344. https://doi.org/10.1029/2021gl094344
- Werdell, P. J., McKinna, L. I. W., Boss, E., Ackleson, S. G., Craig, S. E., Gregg, W. W., et al. (2018). An overview of approaches and challenges for retrieving marine inherent optical properties from ocean color remote sensing. *Progress in Oceanography*, 160, 186–212. https://doi. org/10.1016/j.pocean.2018.01.001
- Wu, Y., Nehrir, A. R., Ren, X., Dickerson, R. R., Huang, J., Stratton, P. R., et al. (2021). Synergistic aircraft and ground observations of transported wildfire smoke and its impact on air quality in New York City during the summer 2018 LISTOS campaign. Science of The Total Environment, 773, 145030. https://doi.org/10.1016/j.scitotenv.2021.145030
- Yang, Z., Demoz, B., Delgado, R., Sullivan, J., Tangborn, A., & Lee, P. (2022). Influence of the transported Canadian wildfire smoke on the ozone and particle pollution over the Mid-Atlantic United States. *Atmospheric Environment*, 273, 118940. https://doi.org/10.1016/j. atmosenv.2022.118940
- Yang, Z., Demoz, B., Delgado, R., Tangborn, A., Lee, P., & Sullivan, J. T. (2022). The dynamical role of the Chesapeake Bay on the local ozone pollution using mesoscale modeling—A case study. *Atmosphere*, 13(5), 641. https://doi.org/10.3390/atmos13050641
- Yorks, J. E., McGill, M. J., Palm, S. P., Hlavka, D. L., Selmer, P. A., Nowottnick, E. P., et al. (2016). An overview of the CATS level 1 processing algorithms and data products. *Geophysical Research Letters*, 43(9), 4632–4639. https://doi.org/10.1002/2016GL068006
- Zhang, J., Ninneman, M., Joseph, E., Schwab, M. J., Shrestha, B., & Schwab, J. J. (2020). Mobile laboratory measurements of high surface ozone levels and spatial heterogeneity during LISTOS 2018: Evidence for sea breeze influence. *Journal of Geophysical Research: Atmospheres*, 125(11), e2019JD031961. https://doi.org/10.1029/2019jd031961



Zhang, W., Guo, J., Miao, Y., Liu, H., Zhang, Y., Li, Z., & Zhai, P. (2016). Planetary boundary layer height from CALIOP compared to radiosonde over China. *Atmospheric Chemistry and Physics*, *16*(15), 9951–9963. https://doi.org/10.5194/acp-16-9951-2016
 Zhang, Y., Sun, K., Gao, Z., Pan, Z., Shook, M. A., & Li, D. (2020). Diurnal climatology of planetary boundary layer height over the contiguous

Zhang, Y., Sun, K., Gao, Z., Pan, Z., Shook, M. A., & Li, D. (2020). Diurnal climatology of planetary boundary layer height over the contiguous United States derived from AMDAR and reanalysis data. *Journal of Geophysical Research: Atmospheres*, 125(20), e2020JD032803. https:// doi.org/10.1029/2020jd032803